Author's response to reviews

Title: Giant Mediastinal Mature Teratoma With Increased Exocrine Pancreatic Activity

Authors:

Franco Stella M.D. (franco.stella@aosp.bo.it)
Fabio Davoli M.D. (fab_78_16l@hotmail.com)
Alessandro Bini Prof. (abini@aosp.bo.it)

Version: 4 Date: 19 January 2011

Author's response to reviews:

I've changed this sentence wrote in the abstract: "Rarely fluid examination from the cystic masses shows a physiological activity, as we described." into "Fluid examination from the cystic masses rarely shows an enzymatic activity, as we described in our report." I think it's better explained.